Author’s response to reviews

Title: Utility of repeat cytological assessment of thyroid nodules initially classified as benign: clinical insights from multidisciplinary care in an Irish tertiary referral centre

Authors:

Nigel Glynn (nigelglynn@physicians.ie)

Mark Hannon (markjhannon2002@yahoo.co.uk)

Sarah Lewis (sarahlewis@rcsi.ie)

Patrick Hillary (patrickhillery@rcsi.ie)

Mohammed Al Mousa (mohammedalmousa@rcsi.ie)

Arnold Hill (adkhill@rcsi.ie)

Frank Keeling (frankkeeling@beaumont.ie)

Martina Morrin (mmorrin@rcsi.ie)

Christopher Thompson (christthompson@beaumont.ie)

Darmaid Smith (diarmuidsmith@beaumont.ie)

Derval Royston (dervalroyston@beaumont.ie)

Mary Leader (maryleader@rcsi.ie)

Amar Agha (amaragha@beaumont.ie)

Version: 3 Date: 17 May 2016

Author’s response to reviews:

Department of Endocrinology
Beaumont Hospital & RCSI Medical School
Beaumont Rd
Dublin 9
17th May 2016
Re: Utility of repeat cytological assessment of thyroid nodules initially classified as benign: clinical insights from multidisciplinary care in an Irish tertiary referral centre.

BEND-D-16-00017R1

Dear Editor and Reviewers,

Thank you for reviewing of our resubmission of the above manuscript.

We are grateful for the favourable response from the Reviewers. In respect of the final editorial requests, please see page 19 & 20 of the revised manuscript.

Yours sincerely,

Dr. Nigel Glynn (on behalf of the authors)